**Author’s response to reviews**

**Title:** The prevalence of stillbirths: a systematic review

**Authors:**

Lale Say (sayl@who.int)
Allan Donner (donner@biostats.uwo.ca)
A. Metin Gulmezoglu (gulmezoglu@who.int)
Monica Taljaard (mtalja@uwo.ca)
Gilda Piaggio (piaggiog@who.int)

**Version:** 4  **Date:** 16 December 2005

**Author’s response to reviews:**

further corrections to the main manuscript and figures were made.